Author's response to reviews

Title: Extra Corporeal Membrane Oxygenation (ECMO) in Three HIV-positive Patients with Acute Respiratory Distress Syndrome

Authors:

francesco giuseppe de rosa (francescogiuseppe.derosa@unito.it)
vito fanelli (vito.fanelli@unito.it)
silvia corcione (corcione.silvia@gmail.com)
rosario urbino (ros.urbino@fastwebnet.it)
chiara bonetto (cbonetto2@cittadellasalute.to.it)
davide ricci (daveri@libero.it)
mauro rinaldi (mauro.rinaldi@unito.it)
giovanni di perri (giovanni.diperri@unito.it)
stefano bonora (stefano.bonora@unito.it)
marco v ranieri (marco.ranieri@unito.it)

Version: 3
Date: 7 April 2014

Author's response to reviews: see over
Dear Editor,

Thank You for the Editorial Comments on our paper entitled “Extra Corporeal Membrane Oxygenation (ECMO) in Three HIV-positive Patients with Acute Respiratory Distress Syndrome”.

Please find hereafter detailed our answers to the Editorial comments:

1. Abstract
   Please clearly provide brief details of what the patient(s) presented with, including the patient's age, sex and ethnic background in the case presentation of your abstract. This has been done.

2. Consent
   Please provide full details of obtained consent. This section should provide a statement to confirm that the patient has given their consent for the Case reports to be published. The editorial office may request copies of the informed consent documentation at any time. We recommend the following wording is used for the consent section: "Written informed consent was obtained from the patient for publication of this Case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal."
   You can use our consent form to obtain consent for publication from the participant(s), or a consent form from your own institution or region if you prefer.
   If the patient has died, then consent for publication must be sought from the next of kin of the patient. If the patient is a minor, or unable to provide consent, then consent must be sought from the parents or legal guardians of the patient. In these cases, the statement in the 'Consent' section of the manuscript should be amended accordingly. This has been done.

3. Acknowledgements
   Please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual
content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible. **No acknowledgements needed**

4. References

Please reformat your references as outlined here: [http://www.biomedcentral.com/bmcanesthesiol/authors/instructions/casereport](http://www.biomedcentral.com/bmcanesthesiol/authors/instructions/casereport)

**This has been done**

5. Tables

Please include remove your tables as additional files and place them at the end of your manuscript as described here: [http://www.biomedcentral.com/bmcanesthesiol/authors/instructions/casereport#preparing-tables](http://www.biomedcentral.com/bmcanesthesiol/authors/instructions/casereport#preparing-tables) **This has been done**

Best regards

Francesco G. De Rosa, MD
Associate Professor,
Infectious Diseases Department,
University of Turin, Italy
Ospedale Amedeo di Savoia
Corso Svizzera 164
10149 Torino
Tel. +39 011 4393979 Fax. +39 011 4393882
e-mail: francescogiuseppe.derosa@unito.it